Author's response to reviews

Title: Treatment of a femoral shaft fracture in a patient with congenital hip disease: a case report

Authors:

George A Tsakotos (gtsakotos@gmail.com)
Stefanos D Koutsostathis (skoutsostathis@gmail.com)
George A Macheras (gmacheras@gmail.com)

Version: 3 Date: 7 May 2010

Author's response to reviews: see over
The following revisions have been made during re-submission:

1. Abstracts in the manuscript and the submission system are identical
2. “et al” has been removed from ref 6. It has been replaced by the co-authors names.
3. Data regarding e-publication in ref 1. has been removed. Publication detail is given (journal, year, issue, pages)
4. All figures have a separate title
5. Highlighting has been removed from the entity of the text
6. Typography has been checked